Author's response to reviews

Title: Fagerstrom Test for Nicotine Dependence vs Heavy Smoking Index in a general population survey.

Authors:

Monica Perez-Rios (monica.perez.rios@usc.es)
Maria Isolina Santiago-Pérez (monica.perez.rios@usc.es)
Begona Alonso (monica.perez.rios@usc.es)
Alberto Malvar (monica.perez.rios@usc.es)
Xurxo Hervada (monica.perez.rios@usc.es)
Jose de Leon (monica.perez.rios@usc.es)

Version: 4 Date: 26 November 2009

Author's response to reviews: see over
Dear Editor:

Thank you very much for your quick reply. Please find enclosed the article “Fagerstrom Test for Nicotine Dependence vs Heavy Smoking Index in a general population survey” which we would like to resubmit with the reviewers’ changes.

We do not need approval of the Galician ethics committee because this is an anonymous research study conducted by telephone and confidentially is fully guaranteed. Informed consent is impossible to achieve because, as we indicated previously, the study was conducted by telephone and participation was anonymous and voluntary.

Sincerely yours
Mónica Pérez-Ríos